Author’s response to reviews

Title: A Rare Case of Avascular Necrosis in Sickle Cell Trait: A Case Report

Authors:

William Sanders (williamsan@pcom.edu)

Version: 2 Date: 27 Nov 2017

Author’s response to reviews:

1. Thanks, I have corrected this.

2. I deemed the data and materials section unnecessary due to the fact that no actual data was gathered from this patient. This case report mainly deals with qualitative findings rather than measurable, quantitative data.

3. Thanks, I have corrected this.

4. Thanks, I have corrected this.

Reviewer Reports:

1. Sorry for the confusion. Actually, this patient simply has a history of asthma, not current symptomatic asthma requiring therapy with steroids, etc. I have included this elaboration in the beginning of the "case presentation" section. Thank you.

2. Because this patient was seen by me in the orthopedic clinic, rather than by another subspecialist such as a cardiologist or hematologist, I gathered the diagnosis from both the orthopedic surgeons word as well as the electronic medical records I obtained from the clinic.

Thank you for your comments. I have uploaded a revision.